Author’s response to reviews

Title: Validation of the Recent Life Changes Questionnaire (RLCQ) for stress measurement among adults residing in urban communities in Pakistan

Authors:

Azmina Artani (azmina.abdul@aku.edu)
Ayeesha Kamal (ayeessa.kamal@aku.edu)
Syed Iqbal Azam (iqbal.azam@aku.edu)
Moiz Artani (moiz.artani@yahoo.com)
Shireen Bhamani (shireen.shehzad@aku.edu)
Mehreen Saif (mehreen.saif1987@yahoo.com)
Fariha Khan (f.afzal.k89@gmail.com)
Nazir Alam (na_hunzai@hotmail.com)

Version: 4 Date: 12 Sep 2019

Author’s response to reviews:

September 12, 2019
Response to Editorial Review
PSYD-18-00171R3

Validation of the Recent Life Changes Questionnaire (RLCQ) for stress measurement among adults residing in urban communities in Pakistan Azmina Artani, MSc Epidemiology and Biostatistics; Ayeesha Kamran Kamal; Syed Iqbal Azam; Moiz Artani; Shireen Shehzad Bhamani, BScN, Msc Epidemiology and Biostatistics; Mehreen Saif; Fariha Afzal Khan; Nazir Alam BMC Psychology

Respected Editor,

Thank you for your time, we have made all the required revisions to the manuscript.

1. Please note that the Consent for publication statement refers to consent obtained to include any individual person’s data in any form (including individual details, images or videos).
Therefore, please state "Not applicable" in this section as no personal information is provided in your manuscript.

It is done.

2. The statement regarding authors’ approval of the final manuscript should instead be included in the Authors’ contributions section of the Declarations.

It is done.

3. The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file, making it available on request to the authors, or depositing it in a public repository and providing the details on how to access it in this section. If you do not wish to share your data, please clearly state this in this section along with a justification. Data availability statements can take one of the following forms (or a combination of more than one if required for multiple datasets):

We have modified this section and used the required statements. (More than one)

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Yes uploaded.

Thank you so much for your time and attention to this manuscript.